Author's response to reviews

Title: Healthcare professionals' intentions to use wiki-based reminders to promote best practices in trauma care: a survey protocol

Authors:

Patrick M Archambault (patrick.m.archambault@gmail.com)
France Légaré (france.legare@mfa.ulaval.ca)
André Lavoie (andre.lavoie.trauma.cha@ssss.gouv.qc.ca)
Marie-Pierre Gagnon (marie-pierre.gagnon@fsi.ulaval.ca)
Jean Lapointe (lcoutjl@videotron.ca)
Sylvie St-Jacques (sylvie.st-jacques@crsfa.ulaval.ca)
Julien Poitras (julienpoitras@videotron.ca)
Karine Aubin (Karine_Aubin@ssss.gouv.qc.ca)
Sylvain Croteau (sylvain.croteau@gmail.com)
Martin Pham-Dinh (martinphamdinh@gmail.com)

Version: 7 Date: 10 June 2010

Author's response to reviews: see over
Wednesday, June 2, 2010

Dear Drs Eccles and Mittman,

I am submitting a third revised version of the study protocol entitled: *Healthcare professionals’ intentions to use wiki-based reminders to promote best practices in trauma care: A survey protocol.*

I must apologize for the many last minute changes, but one of the co-authors just changed her affiliation. Sylvie St-Jacques now works for the Institut national de santé publique. Hence, her affiliation and email has changed. This has changed the order of presentation of the affiliation for the last two co-authors as well (now their affiliation is number 7 rather than number 6).

Sincerely,

__________________________________
Patrick Archambault, M.D., MSc., FRCPC
Corresponding author
Emergency and Critical Care Medicine Specialist
Centre Hospitalier Affilié Universitaire Hôtel-Dieu de Lévis
Post-doctoral Fellow
Research center of the Centre hospitalier universitaire de Québec